Author's response to reviews

Title: Systematic Review of topotecan (Hycamtin) in relapsed small cell lung cancer

Authors:

Rob Riemsma (rob@systematic-reviews.com)
Jean P Simons (jp.simons@live.nl)
Zahid Bashir (zahidbashir@doctors.net.uk)
Caroline L Gooch (caroline.l.gooch@gsk.com)
Jos Kleijnen (jos@systematic-reviews.com)

Version: 6 Date: 17 August 2010

Author's response to reviews:

Dear Editor,

As I mentioned yesterday, it was impossible for me to make changes in the version for which you provided a link in your email (dd 17-08-2010). Therefore, I have made the changes to the latest version I submitted to BMC Cancer on Monday 16 August 2010. As far as I could see the two versions are identical.

I have made the following changes:

- The link for Reference 7 is now:
- This link has been added to reference 7.
- The tables have been removed from the appendix and uploaded as additional files (one for each table A-D).
- All figures are now cited

Minor revisions:

- The appendices have been placed before the ‘competing interests’ and ‘contribution of authors’ sections.
- Figures – Please make these changes for me.
- References – all missing authors have been added.

Best wishes,

Rob Riemsma